Author’s response to reviews

Title: Use of delayed antibiotic prescription in primary care: a cross-sectional study

Authors:
Mariam de la Poza (mariamdelapoza@gmail.com)
Gemma Mas Dalmau (GmasD@santpau.cat)
Ignasi Gich Saladich (IGichs@santpau.cat)
Laura Martínez García (laura.martinez.garcia@cochrane.es)
Carl Llor (carles.llor@gmail.com)
Pablo Alonso-Coello (PAlonso@santpau.cat)

Version: 1 Date: 07 Aug 2018

Author’s response to reviews:

RE: FAMP-D-18-00057 "Use of delayed antibiotic prescription in primary care: a cross-sectional study”

To the Editor:

Thank you for the invitation to re-submit our manuscript “Use of delayed antibiotic prescription in primary care: a cross-sectional study” for consideration of publication.

Below is an itemized summary of the editors’ and reviewers’ comments, followed by our responses and additions to the manuscript. We made the corresponding changes to the manuscript using tracked changes and highlighted to facilitate your review.

Yours sincerely,

Mariam de la Poza Abad